EO-1 Weekly Status Week of October 12 – October 18, 2007

Day of Year 285 - 291
Mission Day 2515 - 2521
Earth Observing-One (EO-1) General:

There were 93 Data Collection Events (DCEs) scheduled this week. 
Instruments underwent decontamination cycling as follows:

The Hyperion hyperspectral imager (HSI) performed a deicing from 10/12/2007 at 22:39z (07-284) to 10/13/2007 at 13:25z (07-285).  

The Advanced Land Imager (ALI) performed an outgassing from 10/12/2007 at 22:47z (07-284) to 10/13/2007 at 13:39z (07-285).

The de-orbit waiver request is proceeding through the approval chain at Headquarters. 

EO-1 Spacecraft Subsystems:

All spacecraft subsystems functioned nominally with the exception of the Command & Data Handling (C&DH) telemetry playback anomaly reported previously.  

Instruments

All instruments operated nominally this week.  

Technology Activities:
In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

· At a request by the Defense Intelligence Agency (DIA), Dan Mandl briefed them on EO-1’s sensor web activity.
· Steve Ungar was asked to be part of the OGC delegation going to the GEOSS summit in South Africa this November.  The OGC will have an exhibit there for the GEOSS Architecture Implementation Pilot and Steve Ungar will be present to discuss EO-1’s role in the Pilot scenarios. 

· Dan Mandl and Vince Ambrosia coordinated potential activities for next summer’s wildfire experiments that will involve the Ames Research Center Ikhana UAS, the JPL Unmanned Aerial Vehicle (UAV) Synthetic Aperture Radar (SAR) experiment, and the EO-1.

Autonomous Sciencecraft Experiment (ASE)

ASE controlled EO-1 all week.

Ground System

Hardware Status

	Hardware
	Status
	Comments

	Primary R/T System
	
	Green
	

	Backup R/T System
	
	Green
	Currently the backup Real-Time system is being utilized to certify the new antennas to be used by the project.  Once all the certifications are complete, the upgrade process will resume.

	bMOC at DataLynx
	
	Yellow
	No change.

	Playback System
	
	Green
	

	Mission Planning
	
	Yellow
	No change.

	Flight Dynamics
	
	Green
	FOT continues to perform re-entry analysis on the spacecraft. New lunar calibration sequence is in the works.  FOT is also designing a new flood lamp calibration sequence.

	DTAS Server
	
	Yellow
	No change.

	ANS
	
	Green
	

	Flight Software Lab
	
	Green
	

	Admin System
	
	Green
	


Operations

New Lunar Calibration Sequence

No change.

New Lamp Calibration Sequence

No change.
Re-Entry Analysis of EO-1

Effort is continuing to refine the EO-1 re-entry analysis.  

New Antenna Additions to EO-1

A connectivity test with SG2 was performed on October 17, 2007 at 14:30 GMT.  Results of the SG2 connectivity test will be posted later.

Use of EO-1 for Solar Dynamics Observatory (SDO) Antenna Tracking Certification
EO-1 continues to provide contacts to the SDOGs antennas. 
New Discrepancy Reports (DR)

There are no new requests this week.

New Enhancement Requests (ER)
There are no new requests this week  

New Operations Database Change Requests (DBCR)

There are no new requests this week.

Ground and Space Network

EOS Data and Operations System (EDOS), GSFC, MD, USA

There are no major problems to report.
McMurdo Ground Station (MGS), McMurdo, Antarctica

There are no major problems to report.

Wallops Ground Station (WGS), Wallops Island, VA, USA

There are no major problems to report.

DataLynx Ground Station (PF1), Poker Flat, AK, USA

There are no major problems to report.

Alaska Ground Station (AGS), Poker Flat, AK, USA 

There are no major problems to report.
Svalbard Ground Station (SGS), Longyearbyen, Norway

There are no major problems to report.

Tracking and Data Relay Satellite System (TDRSS), White Sands, NM, USA

There are no major problems to report.
USGS Multi-Satellite Ground Station (LGS), Sioux Falls, SD, USA
LGS is now being scheduled only for contingencies.

Tasmanian Earth Resources Satellite Station, a.k.a. Hobart (HGS), Hobart, Australia

There are no major problems to report.
Cordoba Ground Station (CGS), Cordoba, Argentina

No passes were scheduled this week. 

Upcoming Events:

· Inclination burn is planned for October 23, 2007.
· Lunar Calibration is scheduled for October 26, 2007

Imagery Status:

Scenes and Engineering Cals planned for week of Oct 12 – Oct 18, 2007                          95
Total scenes and engineering calibrations planned for entire mission – approximately 36,618
Total Scenes:  ALI scenes in the Level 0 archive            33,152 (as of Oct 18, 2007)

                        Hyperion scenes in the Level 0 archive    32,902
Publications and Presentations Status (as of 07-05-07)
404 publications 

284 external presentations

51 articles and press releases
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